Dear Mr. Ma,

Thank you for your letter and for the reviewers' comments concerning our manuscript entitled “Giant cutaneous squamous cell carcinoma of the popliteal fossa skin: A case report” (Manuscript NO.: 72218, Case Report ). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. Revised portion are marked in red in the paper. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

Responds to the reviewer’s comments:

Reviewer #1:
Response to comment: The introduction section of the manuscript is weak, authors are advised to improvise the introduction section.
Response: We have supplemented and improved the introduction of the article.

Reviewer #2:
Response to comment: Author can give brief explain about cutaneous SCC in molecular pathology, about possible ethology, so it can complete and make this manuscript, much better for publication.
Response: We have briefly explained the molecular pathology of cutaneous squamous cell carcinoma in the article.

Reviewer #3:
Response to comment: The grammatical issues should be revised to improve the comprehensibility of this case report.
Response: We have modified the grammar of the article to improve the comprehensibility of the case report.

Special thanks to you for your good comments.

Ke Wang